Author's response to reviews

Title: An automated, broad-based, near real-time public health surveillance system using presentations to hospital Emergency Departments in New South Wales, Australia

Authors:

- David J Muscatello (dmusc@doh.health.nsw.gov.au)
- Tim Churches (tchur@doh.health.nsw.gov.au)
- Jill Kaldor (jkald@doh.health.nsw.gov.au)
- Wei Zheng (wzhen@doh.health.nsw.gov.au)
- Clayton Chiu (cchiu@doh.health.nsw.gov.au)
- Patricia Correll (patriciac@woolcock.org.au)
- Louisa Jorm (ljorm@doh.health.nsw.gov.au)

Version: 5 Date: 7 December 2005

Author's response to reviews:

The modifications to the title and abstract requested by the last reviewer have been made.